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Status 
 Siemens switch was successful. After some tuning, no issue with 

the transfer efficiency in the AGS.  
 AGS cold snake test is ongoing today. Need 8-12 hours behind 

RHIC store(s). No AGS beam activity during this work. 
 AGS and Booster pp setup: starting today if NSRL run finish early 

enough. 
 NSRL runs protons today and tomorrow. 
 For the low intensity store requested by STAR, we need to spend 

time in injector to prepare the beam before Thursday.  
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